Author's response to reviews

Title: A Randomized Controlled Trial Testing the Effectiveness of a Universal School-Based Depression Prevention Program 'Op Volle Kracht' in the Netherlands

Authors:

Yuli R. Tak (y.tak@pwo.ru.nl)
Rinka M. P. Van Zundert (R.vanZundert@pwo.ru.nl)
Rowella C. W. M. Kuijpers (r.kuijpers@pwo.ru.nl)
Boukje S. Van Vlokhoven (bvlokhoven@trimbos.nl)
Hettie F. W. Rensink (hrensink@trimbos.nl)
Rutger C. M. E. Engels (r.engels@pwo.ru.nl)

Version: 2 Date: 1 December 2011

Author's response to reviews:

December 1, 2011

Dear Editor,

Enclosed please find the revised manuscript entitled ‘A Randomized Controlled Trial Testing the Effectiveness of a Universal School-Based Depression Prevention Program ‘Op Volle Kracht’ in the Netherlands’ that my co-authors and I wish to submit for review and possible publication in BMC Public Health. As you have asked, the references are removed from the abstract and a title page is included.

My co-authors and I hope the manuscript will be reviewed and I am looking forward to hearing from you.

Correspondence may be send to Yuli R. Tak.

Sincerely,

Yuli R. Tak, MSc

Radboud University Nijmegen
Behavioural Science Institute
Developmental Psychopathology
Montessorilaan 3, room A 06.18
6525 HR Nijmegen
Email: y.tak@pwo.ru.nl
Phone: +31 243612638